Author's response to reviews

Title: alpha-Glucosidase inhibition, antioxidant and antibacterial activities of two medicinal plants used in Benin as folk medicine

Authors:

Fifa TD Bontho (bothon2006@hotmail.com)
Eric Debiton (eric.debiton@inserm.fr)
Felicien Avlessi (felicien.avlessi@epac.uac.bj)
Christiane Forestier (christiane.forestier@udamail.fr)
Jean-Claude Teulade (j-claude.teulade@u-clermont1.fr)
Dominique KC Sohounhloue (ksohoun@bj.refer.org)

Version: 4 Date: 28 January 2013

Author's response to reviews: see over
Dear Sir,

Please find enclosed our revised manuscript entitled “In vitro biological effects of two anti-diabetic medicinal plants used in Benin as folk medicine” (MS#558482047725059) for publication as a full length article. As recommended, the authors took into consideration all the comments from the reviewer 3.

Hoping that our manuscript will be published in your scientific journal,

Yours Sincerely,

E. Debiton